Author's response to reviews

Title: Prevalence of Cardiovascular Risk Factors across Six African Immigrant Groups in Minnesota

Authors:

Barrett Sewali (sewal026@umn.edu)
Nonyelum Harcourt (harco002@umn.edu)
Susan A Everson-Rose (saer@umn.edu)
Robert E Leduc (robertl@ccbr.umn.edu)
Osman Warfa (warfason@gmail.com)
Michele L Allen (miallen@umn.edu)
Kolawole S Okuyemi (kokuyemi@umn.edu)

Version: 4
Date: 31 January 2015

Author's response to reviews:

UNIVERSITY OF MINNESOTA
Twin Cities Campus Suite 166
Program in Health Disparities Research 717 Delaware Street S.E.
Medical School Minneapolis, MN 55455
Office: 612-625-1654
Fax: 612-626-6782

JAN 31, 2015

BMC Public Health

Journal Editorial Office Biomed Central

Dear Editor-in-Chief:

On behalf of my co-authors and I, I am pleased to resubmit this revised manuscript titled: MS: 1882335941136957 “Prevalence of Cardiovascular (CVD) Risk Factors across Six African Immigrant Groups in Minnesota” for review and consideration by your journal. The current study reports the findings from a cross-sectional survey of 996 African immigrants in Minnesota.

Please find attached reviewers comments and our responses and adjustments as required.

I hereby state that (a) this paper has not been published in whole or in part elsewhere; (b) the paper is not currently being considered for publication
elsewhere; (c) all authors have been personally and actively involved in substantive work leading to the report and will hold themselves jointly and individually responsible for its content; and (d) all relevant ethical safeguards have been met in relation to patient or subject protection.

I look forward to hearing from you. Please address all correspondence on this manuscript to me at the address. Thank you.

Sincerely,

Barrett Sewali MBChB., MPH
sewal026@umn.edu